Author’s response to reviews

Title: The General Practitioner and the diagnosis of occupational diseases: four easy questions

Authors:
Luca Cegolon (mrcegolon@hotmail.com)
John Lange (drcegolon@yahoo.com)
Giuseppe Mastrangelo (giuseppe.mastrangelo@unipd.it)

Version: 2 Date: 14 April 2010

Author’s response to reviews:

Dear BMC Editorial Team,

I hereby re-submit the following paper entitled:” The General Practitioner and the diagnosis of occupational disease”, by Cegolon L et al.

The manuscript has been largely revised and expanded, particularly in the Conclusions. Several extra references have been added as well.

I look forward to hearing from you soon.

Best Regards.

Dr.Luca Cegolon, MD
Corresponding Author